Author's response to reviews

Title: Prevalence and risk factors of major depressive disorder in HIV/AIDS as seen in semi-urban Entebbe district, Uganda

Authors:

   Eugene Kinyanda (Eugene.Kinyanda@mrcuganda.org)
   Susan Hoskins (sjh@ctu.mrc.ac.uk)
   Juliet Nakku (jnakku@yahoo.com)
   Saira Nawaz (nawaz.saira@gmail.com)
   Vikram Patel (vikram.patel@lshtm.ac.uk)

Version: 4 Date: 20 December 2011

Author's response to reviews: see over
To The BMC Series Editorial Production Team

Dear Sir/Madam,

Re: Submission revisions to the manuscript entitled, “Prevalence and risk factors of major depressive disorder in HIV/AIDS as seen in semi-urban Entebbe district, Uganda”

We have addressed the suggested formatting issues as follows:

1) **Issue 1:** Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution  

   *Done*

2) **Issue 2:** Please rename "Methodology" to "Methods"  

   *Done*

3) Issue 3: Tables: Please remove all visible vertical lines from your tables. *I do not see these lines*

4) Issue 4: Figures: The image file should not include the title (e.g. Figure 1... etc.)  

   *Done*

Dr. Eugene Kinyanda MD, PhD

**Correspondent Author**